Author's response to reviews

Title: Chromosome 3q29 deletion, with GI malformation; a case report

Authors: 

MA'IN S MASARWEH (mmasarweh@khcc.jo)

Version: 3 Date: 22 January 2011

Author's response to reviews:

Dear Editor:

I am enclosing the revised manuscript. The additions are written in red font in the main text. I added another reference recommended by the reviewer. I believe that this case is an authentic one and worth reporting because of the unusual GI malformations and the corneal opacities which were not mentioned in any report review. The management of such complex cases are affected by the outcome of these abnormalities and I believe it adds to the medical practice alot and to the diagnostic values as well. The high resolution techniques the reviewer mentioned unfortunately are not available in our country.

What we depend on in such rare cases is to collect all the data published to try to find a sequence to explain the pathogenesis of such rare anomalies. I hope this case will find the acceptance for publication.

best regards
Ma'in Masarweh